Author's response to reviews

Title: Polymorphisms in regulatory regions of Cyclooxygenase-2 gene and breast cancer risk in Brazilians: a case-control study

Authors:

  Diogo N Piranda (diogopiranda@yahoo.com.br)
  Juliana S Festa-Vasconcellos (juliana.s.festa@gmail.com)
  Laura M Amaral (murta.laura@gmail.com)
  Anke Bergmann (abergmann@inca.gov.br)
  Rosane Vianna-Jorge (farmaco@inca.gov.br)

Version: 12 Date: 27 October 2010

Author's response to reviews:

To the BMC Series Editorial Production Team:

The Tables were reformatted as suggested, using MS Word auto-format template: simple 1.

Best regards,

Rosane Vianna Jorge